Author’s response to reviews

Title: Consent to Data Linkage in a Large Online Epidemiological Survey of 18-23 Year Old Australian Women in 2012-13

Authors:

Deirdre Mclaughlin (deirdre.mclaughlin@unimelb.edu.au)

Anna Graves (anna.graves@newcastle.edu.au)

Janni Leung (j.leung1@uq.edu.au)

Jennifer Powers (jenny.powers@newcastle.edu.au)

Version: 3 Date: 20 Nov 2019

Author’s response to reviews:

1. Consent for publication

Please clarify in the "Ethics approval and consent to participate" section of your Declarations that the names ethics committee(s) approved the waiving of formal consent to participate, and accepted that completion of the survey constituted consent.

This has been done.

2. Consent for publication

In your ‘Consent for publication’ section of your declarations, please confirm whether all participants gave written consent for their personal or clinical details along with any identifying images to be published in this study. Consent for publication must be in writing, and cannot be waived if it is necessary. Alternatively, if you believe all identifying information has been appropriately anonymised (e.g. by grouping ages into age ranges) and there are no identifying images used within the manuscript, then consent to publish is not applicable to your manuscript - please write ‘Not Applicable’ in this section if this is the case.

This has been done.
3. Authors’ contributions

Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript in its current state.

The first author has been changed and a Change of Authorship Request form has been completed and submitted by email. The author's contributions have been updated.

4. Role of Funders

Please note the role(s) of the funding body (bodies) in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript, which should be declared in the Funding section.

This has been done.

5. Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

This has been done.